Author’s response to reviews

Title: Identification of cryptolepine metabolites in rat and human hepatocytes and metabolism and pharmacokinetics of cryptolepine in Sprague Dawley rats

Authors:

Arnold Forkuo (forkuo3@gmail.com)
Charles Ansah (cansah.pharm@knust.edu.gh)
David Pearson (david.pearson@novartis.com)
Werner Gertsch (werner.gertsch@novartis.com)
Amanda Cirello (amanda.cirello@novartis.com)
Adam Amaral (adam.amaral@novartis.com)
Jaimie Spear (jaimie.spear@novartis.com)
Colin Wright (c.w.wright@bradford.ac.uk)
Caroline Rynn (caroline.rynn@novartis.com)

Version: 4 Date: 27 Jul 2017

Author’s response to reviews:

RESPONSE TO REVIEWER COMMENTS

COMMENT 1: Please include your statement regarding animal ethics approval in the Ethics approval section of the Declarations.

RESPONSE: Statement on animal ethics has been included in the Ethics approval section of the Declarations.

COMMENT 2: Thank you for providing us with further information regarding change in authorship. We understand it can take time to obtain all necessary signatures, so you may submit this to our editorial office once it has been completed. Please note however, we cannot proceed with acceptance until it has been submitted.
For your reference, the form can be found on the following page:
https://www.biomedcentral.com/submissions/editorial-policies#authorship

RESPONSE: Kindly find attached all necessary signatures of the co-authors. Thank you for the patience. David Pearson and Jaimie Spear had their signatures on separate sheets. Hope that will not be a problem.

COMMENT 3: Please remove your cover letter from the file inventory, as it is no longer needed at this stage of the editorial process.

RESPONSE: This has been done.